Author's response to reviews

Title: An Economic Evaluation of an Augmented Cognitive Behavioural Intervention vs. Computerized Cognitive Training for Post-stroke Depressive Symptoms

Authors:

Mitchel van Eeden (mitchel.vaneeden@maastrichtuniversity.nl)
Joyce A. Kootker (Joyce.Kootker@radboudumc.nl)
Silvia M.A.A. Evers (silvia.evers@maastrichtuniversity.nl)
Caroline M. van Heugten (caroline.vanheugten@maastrichtuniversity.nl)
Ghislaine A.P.G. van Mastrigt (g.vanmastrigt@maastrichtuniversity.nl)

Version: 2
Date: 31 May 2015

Author's response to reviews:

- Line numbers were added to the manuscript
- Author's contributions were added to the manuscript
- Acknowledgements were added to the manuscript